Author’s response to reviews

Title: Solitary fibrous tumor occurring in the parotid gland: a case report

Authors:

Meryem Rais (meryemmrais@gmail.com)
Amine Kessab (aminekessab@gmail.com)
Zahra Sayad (Zahra.sayad999@gmail.com)
Sanae El Mourabit (Sanaeelmourabit@gmail.com)
Redallah Zrarqi (dr.redallah@gmail.com)
Salma Benazzou (benazzousalma@gmail.com)
Malik Boulaadas (mbouladas@yahoo.fr)
Nadia Cherradi (ncherradi2013@gmail.com)

Version: 1 Date: 14 Sep 2017

Author’s response to reviews:

Dear reviewers,

We thank you for the interest shown in the manuscript, and for the opportunity to revise it, your comments were really helpful.

As requested, we have made some changes:

We reported the NAB2-STAT6 gene fusion that has been described in many studies.

We detailed the features associated to prognostication.

We reviewed the quality of written English.

As for the images, we cannot provide better ones this moment. If it is necessary to change the photos, we can supply new ones in October, when the professor that could provide them gets back from vacation.
Thank you for taking the time and energy to help us improve the paper.

Sincerely,

Meryem Rais, MD